Author's response to reviews

Title: Multidisciplinary approach of fused maxillary central incisors: a case report

Authors:

Gilberto Sammartino (gilberto.sammartino@unina.it)
Vincenzo Cerone (v.cerone88@gmail.com)
Roberta Gasparro (roberta.gasparro88@gmail.com)
Francesco Riccitiello (riccitie@unina.it)
Oreste Trosino (trosino@libero.it)

Version: 2 Date: 31 July 2014

Dear Reviewer,

we have modified our submission as you suggested but we cannot include the section of Abbreviation, because there are not abbreviations in the text.

Regarding the section Acknowledgement, there is no one who contributed towards the study and who does not meet the criteria for authorship.